Author's response to reviews

Title: Is the Kidney Disease Quality of Life-36 (KDQOL-36) a valid instrument for Chinese dialysis patients?

Authors:

Susan Ka Yee Chow (susan.chow@polyu.edu.hk)
Bonnie Mee Ling Tam (bonnie.tam@polyu.edu.hk)

Version: 4  Date: 3 September 2014

Author's response to reviews: see over
Dear editor,

Hello. I have revised a paper titled “Is the Kidney Disease Quality of Life-36 (KDQOL-36) a valid instrument for Chinese dialysis patients?” in accordance with the editorial request on September 1, 2014. Below are the revisions on the paper.

1. Page numbers were included in the manuscript
2. Line numbers were included in the paper

The authors’ contributions to the paper are as follows:

- Study design: Susan Chow
- Data collection: Bonnie Tam
- Data analysis: Susan Chow
- Manuscript preparation: Susan Chow & Bonnie Tam

Thank you and I look forward to your favorable reply.

Yours sincerely,

Susan Chow

---------------------------------------------

Susan Chow (Dr)
Assistant Professor
Programme Leader-Master of Nursing
School of Nursing
The Hong Kong Polytechnic University